\begin{table}%t1
\caption{List of the sources analyzed in this paper. Col. I: Object Name; Col. II and III: Right ascension and declination (Equinox 2000); Col. IV: Redshift; Col. V: Seyfert type; Col. VI: Radio class.}\small
\begin{tabular}{lccccl}
\hline\hline
Obj. name            & RA      & Dec       & Redshift&   Obj. type  & Class      \\
\hline
MARK  335               &00 06 19.50&+20 12 11.0 &     0.025&S1.2                 &RQ$^{1}$\\
IRAS 00198-792          &00 21 53.80&--79 10 08.0 &     0.073&S2                   &RQ$^{2}$\\
TON S180                &00 57 20.20&--22 22 56.0 &     0.062&S1n                  &RQ$^{3}$\\
Tololo 109-383          &01 11 27.70&--38 05 01.0 &     0.011&S2                   &RQ$^{4}$\\
MARK 1152               &01 13 50.10&--14 50 45.0 &     0.052&S1.5                 &RQ$^{4}$\\
F   9                   &01 23 45.80&--58 48 21.0 &     0.046&S1.2                 &RQ$^{3}$\\
NGC  526A               &01 23 54.40&--35 03 56.0 &     0.019&S1.9                 &RQ$^{4}$\\
NGC  985                &02 34 37.80&--08 47 15.0 &     0.043&S1.5                 &RQ$^{4}$\\
ESO 198-G24             &02 38 19.70&--52 11 32.0 &     0.045&S1.0                 &RQ?$^{5}$\\
NGC 1052                &02 41 04.80&--08 15 20.8 &     0.005&S2                   &RQ$^{4}$\\
NGC 1068                &02 42 40.70&--00 00 47.0 &     0.003&S2                   &RQ$^{4}$\\
3C 78                   &03 08 26.20&+04 06 39.0 &     0.029&S1                   &RL$^{6}$\\
MARK 1073               &03 15 01.40&+42 02 09.0 &     0.023&S2                   &RQ$^{7}$\\
MARK 609                &03 25 25.36&--06 08 37.9 &     0.034&S1.8                 &RQ$^{4}$\\
3C 88                   &03 27 54.20&+02 33 42.0 &     0.003&S2                   &RQ$^{4}$\\
NGC 1365                &03 33 36.60&--36 08 17.0 &     0.006&S1.8                 &RQ$^{2}$\\
NGC 1358                &03 33 39.70&--05 05 22.0 &     0.013&S2                   &RQ$^{4}$\\
NGC 1386                &03 36 45.40&--35 59 57.0 &     0.002&S2                   &RQ$^{4}$\\
3C 111.0                &04 18 21.30&+38 01 35.0 &     0.048&S1                   &RL$^{8}$\\
3C 120                  &04 33 11.10&+05 21 15.0 &     0.033&S1.5                 &RL$^{4}$\\
UGC  3134               &04 41 48.20&--01 18 07.0 &     0.029&S2                   &RQ$^{9}$\\
ESO 033-G02             &04 55 58.80&--75 32 28.0 &     0.019&S2                   &RQ$^{2}$\\
IRAS 05189-252          &05 21 01.40&--25 21 45.0 &     0.042&S2                   &RQ$^{2}$\\
NGC 2110                &05 52 11.40&--07 27 23.0 &     0.007&S2                   &RQ$^{4}$\\
MCG 8-11-11             &05 54 53.60&+46 26 21.0 &     0.020&S1.5                 &RQ$^{4}$\\
H 0557-385              &05 58 02.10&--38 20 05.0 &     0.034&S1.2                 &RQ$^{10}$\\
MARK    3               &06 15 36.30&+71 02 15.0 &     0.014&S2                   &RQ$^{4}$\\
NGC 2273                &06 50 08.70&+60 50 45.0 &     0.006&S2                   &RQ$^{4}$\\
MARK    6               &06 52 12.30&+74 25 38.0 &     0.019&S1.5                 &RQ$^{4}$\\
MCG-5-18-2              &07 16 31.20&--29 19 28.0 &     0.005&S2                   &RQ$^{11}$\\
MARK 1210               &08 04 05.90&+05 06 50.0 &     0.013&S2                   &RQ\\
MCG-1-24-12             &09 20 46.25&--08 03 22.1 &     0.020&S2                   &RQ\\
MARK  110               &09 25 12.90&+52 17 11.0 &     0.036&S1.5                 &RQ$^{4}$\\
NGC 2992                &09 45 42.00&--14 19 35.0 &     0.008&S1.9                 &RQ$^{4}$\\
MCG-5-23-16             &09 47 40.20&--30 56 54.0 &     0.008&S2                  &RQ$^{12}$\\
M 81                    &09 55 33.17&+69 03 55.0 &     0.0001&S1.8                &RQ\\ 
NGC 3081                &09 59 29.60&--22 49 34.0 &     0.007&S2                  &RQ$^{4}$\\
NGC 3079                &10 01 58.50&+55 40 50.0 &     0.004&S2                   &I$^{4}$\\
NGC 3147                &10 16 53.20&+73 24 02.0 &     0.010&S2                   &RQ$^{13}$\\
NGC 3281                &10 31 52.10&--34 51 13.0 &     0.011&S2                   &RQ$^{4}$\\
RE J1034+393            &10 34 38.60&+39 38 29.0 &     0.042&S1n                  &RQ$^{14}$\\
NGC 3393                &10 48 23.40&--25 09 44.0 &     0.012&S2                   &RQ$^{15}$\\
NGC 3516                &11 06 47.40&+72 34 07.0 &     0.009&S1.5                 &RQ$^{4}$\\
IRAS 11058-113          &11 08 20.30&--11 48 11.0 &     0.054&S2                  &RQ$^{16}$\\
NGC 3627                &11 20 15.03&+12 59 29.6 &     0.002&S2                   &RQ$^{17}$\\
NGC 3660                &11 23 32.20&--08 39 30.0 &     0.011&S2                   &RQ$^{11}$\\
NGC 3783                &11 39 01.80&--37 44 19.0 &     0.009&S1.5                 &RQ$^{4}$\\
NGC 3998                &11 57 56.11&+55 27 12.7 &     0.003&S1                   &RQ$^{4}$\\
NGC 4051                &12 03 09.60&+44 31 53.0 &     0.002&S1n                  &RQ$^{4}$\\
NGC 4151                &12 10 32.50&+39 24 21.0 &     0.003&S1.5                 &RQ$^{4}$\\
MARK 766                &12 18 26.70&+29 48 47.0 &     0.012&S1.5                 &RQ$^{18}$\\
NGC 4258                &12 18 57.50&+47 18 14.0 &     0.002&S2                   &RQ$^{4}$\\
NGC 4261                &12 19 23.22&+05 49 30.8 &     0.007&S2                   &RL$^{19}$\\
MARK  205               &12 21 44.10&+75 18 38.0 &     0.070&S1.0                 &RQ$^{4}$\\
NGC 4388                &12 25 46.70&+12 39 41.0 &     0.008&S2                 &RQ$^{4}$\\
NGC 4507                &12 35 36.50&--39 54 33.0 &     0.012&S2                  &RQ$^{4}$\\
NGC 4579                &12 37 43.52&+11 49 05.5 &     0.005&S1.9                 &RQ$^{4}$\\
NGC 4593                &12 39 39.40&--05 20 39.0 &     0.009&S1.0                 &RQ$^{4}$\\
M 104                   &12 39 59.30&--11 37 23.0 &     0.002&S1.9                 &RQ$^{19}$\\
IC 3639                 &12 40 52.90&--36 45 21.0 &     0.011&S2                  &RQ$^{19}$\\
M 94                    &12 50 53.06&+41 07 13.6 &     0.001&S2                   &RQ$^{17}$\\ 
MARK  231               &12 56 14.20&+56 52 25.0 &     0.041&S1.0                 &I$^{4}$\\
NGC 4826                &12 56 43.76&+21 40 51.9 &     0.001&S2                   &RQ$^{4}$\\
\hline
NGC 4941                &13 04 13.10&--05 33 06.0 &     0.003&S2                   &RQ$^{4}$\\
NGC 4939                &13 04 14.30&--10 20 23.0 &     0.010&S2                   &RQ$^{20}$\\
NGC 4945                &13 05 27.50&--49 28 05.6 &     0.002&S2                   &RQ$^{21}$\\
IRAS 13197-162          &13 22 24.50&--16 43 42.0 &     0.017&S2                   &RQ$^{18}$\\
IRAS 13224-380          &13 25 19.20&--38 24 54.0 &     0.065&S1n                  &RQ$^{14}$\\
Centaurus A             &13 25 27.62&--43 01 08.8 &     0.002&S2                   &RL$^{22}$\\
M 51                    &13 29 52.40&+47 11 41.0 &     0.001&S2                   &RQ$^{23}$\\
MCG-6-30-15             &13 35 53.40&--34 17 48.0 &     0.008&S1.5                 &RQ$^{18}$\\
NGC 5252                &13 38 15.90&+04 32 33.0 &     0.022&S2                   &RQ$^{4}$\\
MARK  266SW             &13 38 17.30&+48 16 34.0 &     0.028&S2                   &RQ$^{4}$\\
IC 4329A                &13 49 19.30&--30 18 34.0 &     0.016&S1.2                 &RQ$^{4}$\\
MARK 463E               &13 56 02.89&+18 22 19.0 &     0.050&S2                   &RQ$^{17}$\\
Circinus                &14 13 09.80&--65 20 17.0 &     0.001&S2                  &RQ$^{4}$\\
NGC 5506                &14 13 14.80&--03 12 26.0 &     0.007&S2                  &RQ$^{4}$\\
NGC 5548                &14 17 59.60&+25 08 13.0 &     0.017&S1.5                 &RQ$^{4}$\\
NGC 5643                &14 32 40.70&--44 10 28.0 &     0.003&S2                   &RQ$^{4}$\\
NGC 5674                &14 33 52.30&+05 27 30.0 &     0.025&S1.9                 &RQ$^{4}$\\
NPM1G-14.0541           &14 37 38.40&--15 00 23.0 &     0.081&S1                   &RQ$^{24}$\\
MARK  478               &14 42 07.50&+35 26 23.0 &     0.077&S1n                  &RQ$^{4}$\\
IRAS 14454-434          &14 48 42.50&--43 55 52.0 &     0.037&S2                  &RQ$^{10}$\\
NGC 5793                &14 59 27.50&--16 41 51.0 &     0.011&S2                   &RL$^{25}$\\
MARK  841               &15 04 01.20&+10 26 16.0 &     0.036&S1.5                 &RQ$^{4}$\\
NGC 6251                &16 32 32.10&+82 32 17.0 &     0.024&S2                   &RL$^{4}$\\
NGC 6212                &16 43 23.10&+39 48 24.0 &     0.030&S1                   &RQ$^{4}$\\
NGC 6240                &16 52 58.90&+02 24 03.0 &     0.024&S2                   &RQ$^{4}$\\
NGC 6300                &17 16 59.20&--62 49 05.0 &     0.003&S2                   &RQ$^{4}$\\
3C 382.0                &18 35 03.40&+32 41 47.0 &     0.059&S1.0                 &RL$^{4}$\\
F   49                  &18 36 58.30&--59 24 09.0 &     0.019&S1.9                 &RQ$^{4}$\\
ESO 103-G35             &18 38 20.50&--65 25 39.0 &     0.013&S1.9                 &RQ$^{10}$\\
3C 390.3                &18 42 09.00&+79 46 17.0 &     0.057&S1.5                 &RL$^{4}$\\
H 1846-786              &18 47 01.70&--78 31 59.0 &     0.074&S1.2                 &RQ?$^{5}$\\
ESO 141-G55             &19 21 14.30&--58 40 13.0 &     0.037&S1.2                 &RQ$^{10}$\\
Cygnus A                &19 59 28.30&+40 44 02.0 &     0.056&S1.9                 &RL$^{26}$\\
IRAS 20051-111          &20 07 51.30&--11 08 34.5 &     0.031&S?                    &RQ$^{5}$\\
IRAS 20210+112          &20 23 25.60&+11 31 34.0 &     0.056&S2                   &RQ$^{24}$\\
MARK  509               &20 44 09.70&--10 43 24.0 &     0.035&S1.5                 &RQ$^{4}$\\
S5 2116+81              &21 14 01.60&+82 04 47.0 &     0.086&S1.0                 &RL$^{27}$\\
PKS 2153-69             &21 57 05.80&--69 41 23.0 &     0.028&S2                   &RL$^{10}$\\
NGC 7172                &22 02 01.90&--31 52 08.0 &     0.008&S2                   &RQ$^{18}$\\
IRAS 22017+031          &22 04 19.20&+03 33 51.0 &     0.066&S2                  &RQ$^{18}$\\
NGC 7213                &22 09 16.25&--47 10 00.0 &     0.06&S1.5                  &RQ$^{18}$\\
3C 445.0                &22 23 49.70&--02 06 13.0 &     0.057&S1.5                 &RL$^{4}$\\
NGC 7314                &22 35 46.10&--26 03 02.0 &     0.005&S1.9                 &RQ$^{4}$\\
AKN  564                &22 42 39.30&+29 43 32.0 &     0.025&S1n                  &RQ$^{14}$\\
MR 2251-178             &22 54 05.90&--17 34 55.0 &     0.068&S1.5                 &RQ$^{4}$\\
NGC 7469                &23 03 15.60&+08 52 26.0 &     0.017&S1.5                 &RQ$^{4}$\\
MARK  926               &23 04 43.50&--08 41 08.0 &     0.047&S1.5                 &RQ$^{4}$\\
NGC 7582                &23 18 23.50&--42 22 14.0 &     0.005&S1                   &RQ$^{4}$\\
NGC~7674                &23 27 56.70&+08 46 44.0 &     0.029&S2                  &RQ$^{4}$\\
NGC 7679                &23 28 46.60&+03 30 42.0 &     0.016&S2                   &RQ$^{7}$\\
\hline
\end{tabular}\medskip
\par
$^{1}$ Barvainis et~al. (2005); $^{2}$ Roy et~al. (\cite{Roy1994}); $^{3}$ Yuan et~al. (\cite{Yuan1998}); $^{4}$ Xu et~al. (\cite{Xu1999}); $^{5}$ No radio-data available in literature; $^{6}$ Yee \& Oke (\cite{Yee1978}); $^{7}$~Marx et~al. (\cite{Marx1994}); $^{8}$ Petre et~al. (\cite{Petre1984}); $^{9}$ Condon et~al. (\cite{Condon2002}); $^{10}$ Brindle et~al. (\cite{Brindle1990}); $^{11}$ Morganti et~al. (\cite{Morganti1999}); $^{12}$ Nagar et~al. (\cite{Nagar1999}); $^{13}$~Ulvestad \& Ho (2001); $^{14}$ Wang et~al. (2004); $^{15}$ Yun et~al. (2001); $^{16}$ Roy et~al. (\cite{Roy1998}); $^{17}$ Condon \& Broderick (\cite{Condon1991}); $^{18}$ Thean et~al. (2000); $^{19}$ Chiaberge et~al. (2002); $^{20}$ Giuricin et~al. (\cite{Giuricin1990}); $^{21}$ Colbert et~al. (\cite{Colbert1996}); $^{22}$ Harris \& Krawczynski (2002); $^{23}$ Rush et~al. (\cite{Rush1996}); $^{24}$ Yun et~al. (2001); $^{25}$ Baan \& Kl{\"o}ckner (2006); $^{26}$ Perley et~al. (\cite{Perley1984}); $^{27}$ Taylor et~al. (\cite{Taylor1996}).
\par
RL, radio loud object; RQ, radio quiet object; I, internediate object.
\end{table}